Sports Inspiration and Inventor Productivity

Jean Canil¹, Douglas Cumming^{2*}, Yudong Liu³, and Chia-Feng (Jeffrey) Yu⁴

^{1,3} Adelaide Business School, University of Adelaide, Australia
 ² College of Business, Florida Atlantic University, USA

⁴ International Business School Suzhou, Xi'an Jiaotong-Liverpool University, China

*Correspondence:

Professor Douglas Cumming. College of Business, Florida Atlantic University. 777 Glades Road, Boca Raton, Florida, 33431 USA. Tel: +1-561-562-0764. Email: <u>cummingd@fau.edu</u>

Emails:

jean.canil@adelaide.edu.au (Jean Canil, PhD and Associate Professor) cummingd@fau.edu (Douglas Cumming, PhD and Professor) yudong.liu@adelaide.edu.au (Yudong Liu, PhD Candidate) chiafeng.yu@xjtlu.edu.cn (Chia-Feng (Jeffrey) Yu, PhD and Senior Associate Professor)

Sports Inspiration and Inventor Productivity

Abstract

This study examines whether sports inspiration influences inventor productivity. We find that long-waited championship of local teams in major U.S. professional sports promotes greater patent volume from local inventors, with heightened forward citations and economic value generated. The effects are more pronounced for non-superstar inventors and underdog victories. Further analysis indicates that inventors pursuing more exploratory innovation strategy and the improvement in the average patent quality are potential mechanisms through which sports inspiration affects inventor productivity. Our results suggest that sports victories can inspire and motivate inventors, leading to increased productivity and impactful innovations. This research provides novel empirical evidence on the nuanced relationship between sports and inventor creativity, highlights non-monetary drivers of inventor productivity, and quantifies the real economic impacts of sports inspiration.

Keywords: Sports Inspiration; Victories; Innovation; Inventor Productivity. **JEL Classification:** L26; L83; M13; G41.

1. Introduction

Innovation is increasingly crucial for national and societal competitiveness, ultimately driving economic growth and success (Solow, 1957; Porter, 1992; Balkin et al., 2000; Ganco, Ziedonis, & Agarwal, 2015; Eisenhardt & Martin 2000). Moreover, innovation enables long-term corporate prospects (Hall, 1993a; Henderson & Cockburn, 1994) and significantly determines a firm's market value (Griliches, 1984; Hall, 1993b). Inventors play a vital role in corporate innovation outcomes (Hall, 2002). Unlike other workers with clear routines, inventors engage in riskier, unpredictable, and complex tasks, requiring significant effort input. Since inventors' effort choices are difficult to observe, it is important to explore the motivations of inventors. What propels these creative minds to push the boundaries of discovery?

The motivations driving corporate inventors are complex and multifaceted. While monetary incentives play a significant role (Lerner & Wulf, 2007; Gao, Hsu & Zhang, 2023), financial rewards alone may not fully capture the essence of inventor motivation. Intriguingly, studies reveal that scientists often accept lower wages to pursue research, effectively "paying to be scientists" (Stern, 2004; Sauermann & Cohen, 2010; Sauermann & Roach, 2014). This phenomenon suggests that inventors' mental states and intrinsic motivations may outweigh purely monetary factors (Osterloh & Frey, 2000; Sauermann & Cohen, 2010; Lam, 2011).

A spectrum of non-monetary motivations has been identified as influential in driving inventor productivity. These include job security (Acharya et al., 2014), intellectual challenge (Sauermann & Cohen, 2010), publishing opportunities (Sauermann & Roach, 2014), autonomy (Sauermann & Cohen, 2010), work environments (Gao et al., 2020), and even residence location (Luo et al., 2022). Given the limitations of financial incentives alone, exploring these non-monetary drivers provides valuable insights into inventor motivation.

Our study aims to expand this understanding by investigating a novel psychological driver: sports inspiration. We posit that the emotional impact of sports victories may

significantly influence inventor productivity and creativity. This choice is grounded in research demonstrating that sports events can elicit profound emotional shifts, affecting individuals' optimism, pessimism, and self-esteem (Depetris-Chauvin, Durante, and Campante, 2020; Jones et al., 2012). These emotional responses often transcend the realm of sports, potentially shaping attitudes, and behaviors across various life aspects, including professional endeavors.

Sports victories have been shown to generate positive moods (Hirt et al., 1992; Jones et al., 2012), which can facilitate creative problem-solving (Isen, Johnson, Mertz, and Robinson, 1985; Fredrickson, 1998; Isen, 1999, 2000; Baas, De Dreu, and Nijstad, 2008; To, Fisher, Ashkanasy, and Rowe, 2012). Furthermore, positivity promotes greater optimism and risk-taking (Tversky and Kahneman, 1973; Isen, Shalker, Clark, and Karp, 1978; Bassi, Colacito, and Fulghieri, 2013) - crucial components of the innovative process (Galasso and Simcoe, 2011; Hirshleifer, Low, and Teoh, 2012; Chen, Podolski, Rhee, and Veeraraghavan, 2014).

By exploring this connection between sports-induced emotions and inventor motivation, our study seeks to bridge these seemingly disparate worlds. We aim to uncover how the emotional responses triggered by sports victories may motivate inventors to engage in more innovative activities, potentially unlocking new pathways to enhance inventor productivity and creativity.

We do so by measuring sports inspiration resulting from major professional sports league championships in the United States¹, including the final championships of the National Football League (NFL), National Basketball League (NBA), Major League Baseball (MLB),

¹ The sports market in the United States is the largest in the world and their major sports events, like NFL, are also characterized by intense fan enthusiasm. In addition, according to the 2016 PwC Sports Outlook (<u>https://www.pwc.com/us/en/industry/entertainment-media/publications/assets/pwc-sports-outlook-2016.pdf</u>), the US sports industry is an important revenue generator in the economy, which was worth \$63.9 billion in 2015, and this will continue to increase.

and National Hockey League (NHL), won by a local team of the city for the first time or at least over ten years since the last victory².

We collect the patent data and inventor information from the Harvard Business School patent inventor database (Li et al., 2004). We restrict our main analysis to inventors affiliated with U.S. publicly listed firms so that we can control for innovation inputs and the characteristics of the firms in which the inventors work³. We define the treated inventors residing in areas affected by sports inspiration, specifically, those living in a city that won a championship of the 'big four' professional sports in the U.S. for the first time or won another championship after a gap of ten years or more. The control group for each sports inspiration event includes inventors from cities that have never won a championship in the big four professional sports won a championship in the big four

Using a large sample of inventor-level data from the Harvard Patent Database from 1975 to 2007, we perform the stacked differences-in-difference estimation to investigate the influence of sports inspiration on inventor productivity. The baseline result shows that the treated inventors produce more patents, receive more forward citations, and have higher economic value relative to the control inventors. Specifically, the treated inventors produce on average 2.1% more patents, and their patents receive on average, 9.8% more forward citations and have on average, 9.8% more economic value than control inventors . Furthermore, our findings still hold after several robustness tests, including matching employing the propensity score matching (PSM) method (Rosenbaum and Rubin, 1983). We show that prior to the event, both treatment and control groups exhibited no significant difference in inventor innovative activities, but after the event, there was a significant increase in inventor productivity for

² Based on a survey by Gallup in the U.S. (https://news.gallup.com/poll/4735/sports.aspx), over 60% of respondents describe themselves as sports fans in general, and around 65% state that their favorite sports are football, basketball, basketball, and ice hockey.

³ In the last section, we also focus on all inventor and our results still hold.

treatment inventors, which also confirms the parallel trend assumption underlying our empirical design. Our tests show that the pre-treatment trends in inventor productivity are indistinguishable between these two groups, with most of the effect occurring after the sports inspiration event, suggesting a causal effect.

To examine whether sports inspiration has varying effects on inventor productivity, we perform a series of additional tests. First, we show that sports inspiration has a greater effect on the innovative productivity of non-superstar inventors than the productivity of superstar⁴ inventors. We also show that the sports inspiration events caused by an underdog win have a more significant effect on inventor productivity than that caused by victory from a non-underdog team.

Furthermore, we explore potential channels through which sports inspiration may influence inventor productivity. First, we assess whether treated inventors alter their innovation strategies following inspirational events, specifically producing more exploratory and risktaking patents versus exploitative patents. We find treated inventors pursue more exploratory patents. Furthermore, we examine cognitive impacts, which could enable beneficial R&D decisions and higher-quality patents. We find average patent quality increases for treated versus control inventors' post-inspiration.

Our study contributes to three streams of literature. First, we add to the literature on sports outcomes' effects on economic phenomena by extending beyond short-term stock market studies (e.g., Edmans et al., 2007) to analyze long-term innovation impacts. This approach bridges the gap between immediate market reactions and sustained economic productivity. Second, we present large-scale evidence on how sports-induced mood influences creativity, adding nuance to the conflicting results of small-scale psychology experiments

⁴ Following Zacchia (2018), we define inventors as superstar inventors as those in the top 5% of the distribution of patents that the inventor filed by event

(Davis, 2009). Our comprehensive dataset of inventor productivity offers a more robust perspective on this relationship, helping to reconcile previous contradictions in understanding the interplay between emotional states and cognitive functions in real-world contexts. Finally, we expand the scope of innovation research by focusing on inventors, a group that has received less attention compared to executives, boards, and shareholders. While existing literature has examined how behavioral traits of top management and governance structures affect innovation (Galasso and Simcoe, 2011; Hirshleifer, et al., 2012; Ederer and Manso, 2011, 2013; Balsmeier, Fleming, and Manso, 2017)⁵, we uniquely demonstrate that inventors' productivity, shaped by external factors like sports outcomes, can significantly influence both innovation output and market value. This perspective adds a new dimension to understanding the drivers of corporate innovation, highlighting the importance of considering broader psychological and environmental factors affecting those directly responsible for innovative activities. Overall, this work introduces a novel psychological factor driving inventor productivity, bridging the gaps between sports psychology, creativity research, and innovation studies, and provides valuable insights for both academic research and corporate innovation management. The rest of this paper is organized as follows. Section 2 reviews relevant literature and develops the hypothesis. Section 3 describes the data and variable measurements. Section 4 describes the empirical design, including the classification of treatment groups and control groups and the stacked DID model. Section 5 presents the results of the baseline analysis, robustness, a series of additional analyses, and potential mechanism analysis. Section 6 concludes.

2. Literature Review and Hypothesis Development

⁵ It is worth noting that some research has explored how incentives and compensation plans for non-executive employees can positively influence a firm's innovative activities. Examples include studies by Lerner and Wulf (2007), Acharya, Baghai, and Subramanian (2014), Chang, Fu, Low, and Zhang (2015), and Chen, Chen, Hsu, and Podolski (2016). However, our study differs by focusing on external psychological factors rather than internal organizational policies.

2.1.1 The Importance of Inventor

Innovation plays an increasingly crucial role in national and societal competitiveness, ultimately driving economic growth and success (Solow, 1957; Porter, 1992; Balkin et al., 2000; Ganco, Ziedonis, and Agarwal, 2015; Eisenhardt and Martin 2000). It has been shown to be essential for the long-term success of corporations (Hall, 1993a; Henderson and Cockburn, 1994) and significantly determines a firm's market value (Griliches, 1984; Hall, 1993b). Corporate innovative success lies in the hands of inventors working on research and development (R&D) projects (Hall, 2002), with their creativity being "the key ingredient for job creation, innovation and trade" (United Nations Conference on Trade and Development, 2010). Unlike routine workers with clear goals (e.g., production workers, salespersons, and administration staff), inventors' tasks are risky, unpredictable, and complex. These job features necessitate substantial autonomy in inventors' activities, making it difficult to monitor their performance. Consequently, the voluntary efforts made by inventors are essential to maintaining a productive inventor workforce. Understanding the motivations of individual inventors is therefore a critical issue, as it directly impacts the innovative capacity of firms and, by extension, entire economies.

2.1.2 Non-money Incentive of Inventor Productivity

Prior studies document that monetary incentives play a significant role in motivating corporate inventors (Lerner and Wulf, 2007; Gao, Hsu & Zhang, 2023). The study by Lerner and Wulf (2007) shows that offering long-term incentives to corporate research and development leaders, such as stock options and restricted stock, leads to an increase in the number of times their patents are cited. Gao et al. (2023) examine the role of pay transparency

in the productivity of firms' and inventors' innovation activities using the staggered adoption of the state-level pay secrecy laws and find that there is a significant increase in inventor productivity of firms located in states that have passed such laws relative to firms elsewhere.

However, monetary incentives may not be sufficient. The economics literature documents that scientists "pay to be scientists" because they are willing to sacrifice wages for the opportunity to do research (Stern, 2004; Sauermann and Cohen, 2010; Sauermann and Roach, 2014). Other non-monetary incentives include job security (e.g., Acharya, Baghai, and Subramanian, 2014), intellectual challenge (Sauermann and Cohen, 2010), academic publication opportunities (Sauermann and Roach, 2014), and autonomy (Sauermann and Cohen, 2010). Recently, Gao et al. (2020) identified a positive causal effect of healthy working environments on corporate innovation at the corporate inventor's level, using the staggered passage of U.S. state-level laws that ban smoking in workplaces. In addition, Luo et al. (2022) find that there is a positive relationship between air quality and the productivity of patent inventors using the NOx budget program (NBP) as a quasi-natural experiment. Others argue that the mental state of inventors is more important than monetary incentives (Osterloh and Frey, 2000; Sauermann and Cohen, 2010; Lam, 2011). Clearly, understanding the effect of non-monetary incentives on inventors is meaningful.

2.2 The Impact of Sports Event Outcomes

The following review explores a plethora of studies investigating how individuals emotionally respond to sports results and how these responses affect behavior, including mood, self-esteem, and economic decisions that extend beyond the confines of sports arenas.

2.2.1 Emotional Responses to Sports Event Outcomes

A substantial body of research consistently demonstrates the profound emotional impact of sports events on individuals (Schwarz et al., 1987; Hirt et al., 1992; Schweitzer et al., 1992; Wann et al., 1994; Jones et al., 2012; Ge et al., 2021; Cardazzi et al., 2022). Favorable outcomes, such as victories or strong performances by one's favorite team, tend to evoke positive emotions and elevate the mood of sports enthusiasts. Conversely, disappointing results are invariably associated with negative emotions. The seminal work of Wann et al. (1994) emphasizes that these emotional reactions often extend well beyond the immediate aftermath of the sporting event, significantly impacting individuals' self-esteem and overall life satisfaction.

Further, Hirt et al. (1992) reveal a noteworthy improvement in the academic performance of college students after witnessing their team win a basketball match, highlighting the spill-over effects of sports outcomes into non-sport-related domains. Similarly, Schwarz et al. (1987) document that Germany's World Cup match outcome in 1982 induced notable changes in subjects' well-being and perceptions of national issues. In a parallel vein, Schweitzer et al. (1992) demonstrate that students supporting the winning team in a televised American football game exhibited lower evaluations of the probability of a 1990 war in Iraq and its potential casualties than fans of the losing team.

Jones et al. (2012), analyzing survey data from English and Spanish soccer fans during the 2010 World Cup, found enduring positive emotional experiences associated with group success, which persisted longer than the negative emotional experiences linked to group failure. Similarly, Ge et al. (2021) present evidence of a notable increase in thefts and robberies in San Paulo, Brazil, following football matches, with upset losses and derby games eliciting particularly pronounced effects. Additionally, Cardazzi et al. (2022) establish a compelling correlation between unexpected losses by the local NBA team and an increase in male-onfemale in-home violence. Recent research by Depetris-Chauvin, Durante, and Campante (2020) examines the role of shared collective experiences in building national identity by studying the impact of national football teams' victories in sub-Saharan Africa. Their findings reveal that individuals surveyed in the days after an important victory of their country's national team exhibit a 37 percent lower likelihood of primarily identifying with their ethnic group and a 30 percent increase in trust in other ethnicities compared to those interviewed just before. Crucially, national team achievements also reduce violence, with countries that (barely) qualified for the Africa Cup of Nations experiencing 9 percent fewer civil conflict episodes in the following months than countries that (barely) did not.

2.2.2 Economic Implications of Emotional Responses to Sports Outcomes

The emotional responses elicited by sports events have reverberations in the realm of economic activities, as evidenced by Arkes et al. (1988), who observed an increase in Ohio State lottery ticket sales following the victory of the Ohio State University football team. In the financial markets, Ashton et al. (2003), Edmans et al. (2007), Kaplanski and Levy (2010a), Chang et al. (2012), and Pantzalis et al. (2014) assert that sports results can significantly impact stock returns. Furthermore, Akhigbe et al. (2017) investigated the influence of predictable sports sentiment on local trading activities and found statistically significant increased trading before games. Using household-level data, Kaplanski et al. (2015) provide compelling evidence that sports results and general feelings substantially affect stock market return expectations, with a strong positive correlation between the success of an individual's favorite sports teams and their expectations.

2.3 Hypothesis Development: Sports Inspiration and Inventor Productivity

In this study, we investigate the influence of sports inspiration on inventor innovation productivity. We define sports inspiration as exogenous shocks stemming from championship victories in major professional sports leagues in the United States, including the NBA, NFL, NHL, and MLB, occurring at the city level over a ten-year period. Distinguishing itself from mere sports sentiments utilized in prior research, sports inspiration events are characterized by their precision, specificity, and the enduring impact they impart upon individuals.

There is a large body of evidence that shows that sports victories can generate positive moods (Hirt et al. (1992); Jones et al. (2012)), which can facilitate creative problem solving, cognitive elements (Isen, Johnson, Mertz, and Robinson, 1985; Fredrickson, 1998; Isen, 1999, 2000; Baas, De Dreu, and Nijstad, 2008; To, Fisher, Ashkanasy, and Rowe, 2012). Furthermore, these positive mood increases confidence in one's abilities, thus inducing individuals to show more initiative at work and pursue challenges with greater persistence (Kavanagh and Bower, 1985).

In addition, a positive mood has been shown to promote greater optimism and risktaking (Tversky and Kahneman, 1973; Isen, Shalker, Clark, and Karp, 1978; Bassi, Colacito, and Fulghieri, 2013), which are important components of the innovative process (Galasso and Simcoe, 2011; Hirshleifer, Low, and Teoh, 2012; Chen, Podolski, Rhee, and Veeraraghavan, 2014).

Building upon psychological insights and behavioral theories, we posit the following hypothesis:

Hypothesis 1: Sports inspiration promotes inventor innovation productivity.

3. Data and Sample

To empirically examine the effect of sports inspiration on inventor productivity, we collected patent data and inventor information from the HBS Patent Inventor Database (Li et al., 2014). This database records every patent granted by the U.S. Patent and Trademark Office (USPTO) from 1975 to 2010. From the database, we obtained detailed information about the

inventors of each patent, including their names, cities of residence, and zip codes. Using a disambiguation algorithm, the database assigns each inventor a unique identifier, which enables us to track their innovation records, along with their accurate residential location. To account for heterogeneity among inventors, we also control for innovation inputs and the characteristics of the firms for which the inventors work. As such, we restrict our main analysis to inventors affiliated with U.S. publicly listed firms (Chen et al., 2022; Luo et al., 2022; Fich et al., 2023). We use the HBS Patent Inventor Database to obtain corresponding citation data as well. Patent inventors are matched with U.S. publicly listed firms based on patent data from Kogan et al. (2017), who provide the Center for Research in Security Prices (CRSP) firm identifier for each patent. We collect financial data for these publicly listed firms from Compustat.

Because an inventor is listed in the HBS Patent Inventor Database only when they file a patent, our original sample comprises inventor-year observations in which inventors filed at least one patent in a given year. We identify the first and last year in which an inventor files a patent in the Patent Inventor Database (Baghai et al., 2019). We then assign a value of zero to the inventor's innovation output variables for all years with no patent record. In this way, we create consecutive time series data for all inventors⁶. As an inventor's residential information is available only when they file a patent, we assign the inventor's most recent residential information to the years for which there is no patent record (Hombert and Matray, 2017). We use the application year (i.e., the filing year) of a patent as the time marker in our empirical tests, since the application year should be closest to the time when the new technology appeared (Hall and Ziedonis, 2001). Given an application-approval lag of two to three years (Hall and Ziedonis, 2001), we exclude the final three years (2008-2010) from our analysis. Therefore,

⁶ We use patent-based measures to gauge inventor performance with caution and acknowledge their shortcomings (Lerner and Seru, 2022; Jaffe and de Rassenfosse, 2017). For example, inventors might choose not to patent their inventions due to concerns such as information leakage. Therefore, patent-based measures could be noisy measures of inventor performance, especially in industries with low patent propensity (Cohen, Nelson, and Walsh, 2000; Hall, Helmers, Rogers, and Sena, 2014).

our sample period starts in 1975 and ends in 2007. The inventor-level data constitute the dataset used in our baseline analysis.

3.1 The Measurement of Sports Inspiration

To measure sports inspiration, we hand collect the championship data of four major sports leagues (NFL, MBA, MLB, and NHL) in the U.S. We define an external event, or exogenous shock when a city wins its first sports championship or has not achieved a championship victory in over a decade. We use this event to gauge its impact on sports inspiration. For example, in 1999, San Antonio Spurs won its first NBA championship, the city's first championship of the four major sports leagues. Similarly, for Boston in Massachusetts, the Boston Red Sox won the 2004 MLB championship, over ten years since it won the last championship (Boston Celtics won the NBA championship in 1986). These constitute our sports inspiration events. More detailed information can be found in Figure 1.

[Insert Figure 1 here]

MLB, NHL, NFL, and NBA were founded in 1903, 1917, 1922, and 1946, respectively. Based on the definition of sports inspiration we propose above, the first exogenous shock to measure sports inspiration is Boston Americans (changed to Boston Socks) when it won its first MLB championship in 1903. In this paper, our sample period is from 1975 to 2007. Thus, in our project, the first sports inspiration event is the NBA championship won by Portland Trail Blazers for Portland in 1977, which is the city's first championship of the four major sports leagues, and the last sports inspiration shock is the Indianapolis Colts who won the NFL champion for Indianapolis in 2007.

3.2 The Measurement of Inventor Productivity

We use three measures of an inventor's innovation output, based on newly filed patents by the inventor during the filing year that are eventually granted. The first measure is the number of patents, calculated as the total number of newly filed patents. The second measure is the number of citations, which is determined by the sum of all forward citations received by these newly filed patents. Studies indicate that forward citations received by a patent reflect the patent's scientific value, with the expectation that more ground-breaking patents will receive a greater number of citations than those that are less innovative (Hall et al., 2002; Hall et al., 2005; Aghion et al., 2013). The third measure is total patent value, calculated as the sum of the economic value of all newly filed patents. According to Kogan et al. (2017), this measure is calculated as the increase in the market value of the firm (after adjusting for benchmark returns) within a three-day window following the patent grant announcement. As the three innovation output measures are highly skewed, we take the natural logarithm of 1 plus the number of patents, number of citations, and total patent value separately, and use these logtransformed measures in our analysis.

Following Moretti (2021), we adjust the number of patents, citations and values attributed to an inventor in a given year based on the number of inventors listed for a specific patent. If a patent has multiple inventors, we assign equally weighted fractions of the patent and its citations to each inventor. For instance, if a patent lists four inventors, each inventor is credited with one-quarter of a patent and one-quarter of all subsequent citations.

3.3 The Measurement of Firm Characteristics

Our analysis is based on a sample of inventors affiliated with U.S. public firms, which enables us to control for a set of firm-level variables in our analysis. First, as large firms usually generate more patents and citations than small firms (Hall and Ziedonis, 2001), we include firm size (*Asset*), defined as the natural logarithm of total assets, in the control set. To control for

the firm's innovation inputs, we include R&D expenses (R&D), defined as R&D expenditure scaled by total assets. Following prior studies (e.g., Hirshleifer et al., 2012), we set R&D for observations with missing R&D information in Compustat to 0. We also control for firms' capital investment (CapEx), defined as capital expenditures scaled by total assets, return on assets (ROA), defined as earnings before interest and tax divided by total assets, cash holding (Cash), defined as cash and short-term investments scaled by total assets, leverage (Leverage), defined as the book value of debt scaled by total assets, and Tobin's Q (Tobin's Q), defined as market value of equity plus book value of total assets minus book value of equity minus balance sheet deferred taxes, normalized by book value of total assets. Finally, we control for the effect of the life cycle of firms by including firm age (*Firm_Age*), defined as the natural logarithm of 1 plus the number of years elapsed since the first year in which that firm appeared in the Compustat database. As information about inventor characteristics is limited, the only inventor-year variable we control for is the inventor tenure (Tenure), defined as the natural logarithm of one plus the number of years between the year in which the inventor entered the patent database and the observation year. Detailed descriptions of all variables are provided in Appendix A.

3.4 Descriptive Statistics

Table 1 reports the descriptive statistics of the key variables in the empirical analysis. The table shows that the means of *LnPat*, *LnCit*, and *LnVal* are 0.244, 1.022, and 0.857 respectively. With regard to the control variables, the mean of the *Asset* is 8.652 and the mean of the *R&D* is 0.08. The average value of *ROA*, *Leverage*, *CapEx*, *Cash*, *Tobin's Q*, and *Firm_Age* for the firms in our sample is 0.134, 0.185, 0.066, 0.186, 2.615, and 3.164, respectively. In addition, the mean of *Tenure* is 1.907, corresponding to 5.73 years, which suggests relatively long time series data for the average inventor.

[Insert Table 1 here]

4. Empirical Methodology

4.1 Defining Treatment and Control Groups

Our treatment group consists of inventors residing in cities affected by sports inspiration, namely, those living in a city that won a championship of the 'big four' professional sports in the U.S. for the first time or won another championship after a gap of ten years or more. We examine the changes in both the quantity and quality of inventor productivity over a seven-year window surrounding the event (the event year, three years before, and three years after). Our control group for each sports inspiration event includes inventors from cities that have never won a championship in the big four professional sports in the U.S., from the same state as the treated cities.

In many cases, both the inventor's residential address and the assignee's address (typically the company that initially owns the patent) are available. Following Moretti (2021), we do not use the assignee address because it may not reflect the actual location where the research was conducted; often, it is merely the address of the corporate headquarters, not the R&D facility.⁷

⁷ In our sample, around 46% of inventors had a residential state different from the headquarters of their employing firms from 1975 to 2007. Additionally, from 1993 to 2007, around 96% of inventors had a residential city different from the headquarters of their employing firms.

4.2 Stacked DID Model⁸

To investigate how sports inspiration affects inventor productivity, we perform the following difference-in-differences (DID) analysis using a stacked sample:

$$Y_{i,t+1,j} = \beta_1 Treat_{ij} * Post_{tj} + \Sigma Controls + Inventor_j + Year_j + City_j + Firm_j + \epsilon_{itj}.$$
(1)

The dependent variable $Y_{i,t+1,j}$ represents the quantity (*LnPat*) and quality (*LnCit*) index of inventor productivity for inventor i in year t + 1 for the event j. *Treat* is an indicator variable that equals one (zero) for the treatment (control) group. To conduct a difference-indifferences analysis, we also define an indicator variable, *Post*, which equals one for years after the sports inspiration event. *Inventor_j*, *Year_j*, *City_j*, and *Firm_j* are the inventor-, year-, city-, and firm-fixed effects for each event j, respectively. ϵ_{itj} is the residual of the model. All of the t-statistics are on an adjusted basis, clustered⁹ by event*city (White 1980; Petersen 2008; Cengiz, Dube, Lindner, and Zipperer, 2019).

Our key interest is the estimate of the coefficient β_1 , which captures the effects of the influence of sports inspiration on inventor productivity. If sports inspiration promotes inventor productivity, we should observe positive and significant coefficient estimates on $Treat_{ij} * Post_{tj}$.

5. Empirical Results

⁸ Traditional staggered DID analysis may generate biased estimates because of negative weights in the presence of heterogeneous treatments effects, which can be alleviated by the stacked DID identification strategy (e.g., Cengiz, Dube, Lindner, and Zipperer, 2019; Baker, Larcker, and Wang, 2022). So, we use a stacked DID model to investigate the sports inspiration on inventor productivity, and treatment and control cities are defined event by event.

⁹ Considering the intra-cluster correlation, many patents are assigned at the team/firm level, which means there is a strong (mechanical) correlation in patent output across inventors within a team-firm. This paper focuses on the individual inventor level; therefore, we re-run the baseline analysis clustered at the event-firm level to rule out this concern and our results still hold. Details can be seen in Appendix B.

5.1 Baseline and Parallel Trend

We use the stacked DID model to investigate the influence of sports inspiration on inventor productivity. In Table 2, we examine the effect of sports inspiration on inventor productivity using the regression model (1). Column (1), Column (3), and Column (5) show the estimation of the impact of sports inspiration on *LnPat*, *LnCit*, and *LnVal*, respectively, with only fixed effects. The coefficient of *Treat*Post* is positive and significant at 1% level for all three results. In Column (2), Column (4), and Column (6), we include control variables in our regression analysis. The coefficient of *Treat*Post* is positive and significant at 1% level for all three outcomes suggesting that inventors not only generate more patents but also generate more highly qualified patents after sports inspiration, relative to controlled inventors. In terms of economic magnitude¹⁰, treated inventors file 8.6% more patents, and their patents receive more 9.59% more forward citations, and create 11.44% more economic value relative to controlled inventors after sports inspiration. Therefore, the effect of sports inspiration on inventor innovation is also economically significant.¹¹

[Insert Table 2 here]

¹⁰ As for the calculation economic significance of dummy variable *Treat*Post*, we follow Mitton (2021) using the formula $E_{\frac{1}{y}} = |\frac{b}{y}|$. b represents the coefficient of the *Treat*Post* and \overline{y} represents the mean value of the dependent variable, here is the mean value *LnPat*, *LnCit*, and *LnVal*. For example, the economic magnitude of the influence of sports inspiration on the number of newly filed patents in Column (2) Table 2, we calculate it using the value of the coefficient of the *Treat*Post* 0.021 and value of the mean value of *LnPat* 0.244 following the formula $E_{\frac{1}{y}} = |\frac{b}{\overline{y}}|$. We replace b with 0.021 and \overline{y} with 0.244 and then obtain the value of $E_{\frac{1}{y}}$ 8.6%.

¹¹ The economic significance of our study is reasonable, in particular when compared with that document in prior studies. For example, Chen et al. (2014) find that a one standard deviation increases in the Catholics-to-Protestant ratio, an indicator of local gambling preference, is associated with a rise of approximately 8% of a firm's patent counts. Furthermore, Galasso and Simcoe (2011) note that the presence of an overconfident CEO is associated with 25% to 30% increase in citation-weighted patent counts. Hirshleifer et al. (2012) reports a 28% increase in patent counts for firms led by overconfident CEOs.

To ensure the validity of the parallel trend assumption for our DID analysis, we examine the dynamic effect of sports inspiration on inventor innovation in Table 3. Specifically, we select the third year before the sports inspiration year as the benchmark year. We replace *Post* in Model (1) with five, time dummies¹², including *Before²*, *Before¹*, *After¹*, *After²* and *After³*. *Before²* equals one for observations in the second year before the sports inspiration event. *Before¹* equals one for observations in the first year before the sports inspiration event. *After¹* equals one for observations in the first year after the sports inspiration event. *After²* equals one for observations in the first year after the sports inspiration event. *After³* equals one for observations in the sports inspiration event. *After³* equals one for observations in the third year after the sports inspiration event.

We include the interaction terms between *Treat* and these five dummy variables on our baseline regression along with our control variables. In Column (1), the coefficients for Treat**Before*² and *Treat***Before*¹ are indistinguishable from zero, indicating no pre-existing trends in local inventor innovation from the quantity perspective before the sports inspiration. In contrast, the coefficients of *Treat***After*¹ are significantly positive at 5% level. The pattern in our dynamic analysis demonstrates that the parallel trend assumption holds. In Column (2) for the quality index of the inventor innovation, the coefficients for Treat**Before*² and *Treat***Before*¹ are significantly positive at 1% level. In contrast, the coefficients of *Treat***After*³ are significantly positive at 1% level. Similarly, in Column (3) for the quality index of the inventor innovation, the coefficients for Treat**Before*² and *Treat***Before*¹ are indistinguishable from 2, indicating no pre-existing trends in local inventor innovation, the coefficients of *Treat***After*³ are significantly positive at 1% level. Similarly, in Column (3) for the quality index of the inventor innovation, the coefficients for Treat**Before*² and *Treat***Before*¹ are indistinguishable from zero, indicating no pre-existing trends in local inventor innovation from the quality positive at 1% level. Similarly, in Column (3) for the quality index of the inventor innovation, the coefficients for Treat**Before*² and *Treat***Before*¹ are indistinguishable from zero, indicating no pre-existing trends in local inventor innovation from the quality perspective before the sports inspiration. In contrast, the

¹² We drop the event year in our parallel trend test to avoid noise, as the World Series of MLB usually takes place in October, the NBA Finals generally occur in June, the NHL Finals also generally occur in June, and the NFL Finals typically take place in January or February. Our baseline analysis still holds even when we drop the event year observations, and our parallel trend test still passes even when we keep the event year observations.

coefficients of *Treat*After¹*, *Treat*After²*, and *Treat*After³* are significantly positive at 5% level. The pattern in our dynamic analysis demonstrates that the parallel trend assumption holds.

[Insert Table 3 here]

5.2 Robustness Tests

To check the validity of our baseline results and exclude alternative explanations, we conduct a number of robustness tests. Table 4 shows the robustness results of our baseline results on the effect of sports inspiration on local inventor innovation productivity. First, we examine whether our results are sensitive to an alternative sample. In our baseline results, we assign a value of zero to the innovation output variables for all years with no patent record and between the first and last years in which the inventor filed a patent. To ensure that our results are not driven by this treatment, we exclude the inventor who did not file any patent during the event window and re-estimate the baseline analysis on this sample. We obtain consistent findings, which are presented from Column (1) to Column (3) in Table 4. Similarly, we exclude inventors who are employed by the financial and utility firms. The results are shown in Column (4) and Column (6), showing consistent results.

Thirdly, we consider that the control group not only includes the inventors living in zip code areas from the same state as the treatment zip code areas but also includes all inventors living in the zip code areas that never won a championship of big four professional sports in the U.S. For example, in our baseline regression, for the sports inspiration event of Boston winning the 2004 MLB championship, the control group includes the inventors who live in the zip code areas in Massachusetts which never won a champion of the big four professional sports. In this robustness section, the control group includes all the inventors who live in the

zip code areas in the whole United States which never won a champion of the big four professional sports. The results are reported in Column (7) and Column (9) in Table 4 and are consistent with our baseline results.

To alleviate further concerns that our findings may be driven by pre-event differences between the treatment and control groups, we conduct a propensity score matching (Rosenbaum and Rubin, 1983) and then re-run our regression model (1) on the matched sample. Specifically, for each treatment inventor in an event, we select a matched control inventor based on a propensity score from the logit model. We conduct the matching process 30 times for the 30 events used in our previous regression analysis. In the logit model, the dependent variable is the *Treat* dummy, and the matching variables include *Asset* and *Ln_tenure* (Chen et. al., 2022). Our logit model also controls for industry-fixed effects. We define industries based on the classification of Fama-French 48 countries. To maintain the statistical independence of our tests, we implement a nearest neighbor matching (NNM) algorithm without replacement and match inventors with a similar propensity score. The NNM algorithm uses the distance between covariate patterns to define the "closest" neighbor. Column (10) and Column (12) of Table 4 show the regression results with the propensity score matched sample. The coefficients of *Treat***Post* are 0.028 (standard errors = 0.009), 0.111 (standard errors = 0.036), and 0.081 (standard errors = 0.032) in Column (10), Column (11), and Column (12) respectively, which are positive and significant. Thus, we continue to find significant evidence that sports inspiration affects treated inventor innovation productivity when controlling for potential similarities in the pre-event period.

[Insert Table 4 here]

5.3 Subsample Analysis

In this section, we will do a series of subsample analyses to explore the varying effects of sports inspiration on local inventor innovation productivity.

5.3.1 Inventor Talent

Chen et al. (2022) and Graff Ziven and Neidell (2012) show that more experienced inventors such as superstar inventors are better at adapting to external factors such as air pollution, thus it is possible that these same inventors are less likely to be influenced by sports inspiration. Following Zacchia (2018), we define inventors as superstar inventors as those in the top 5% of the distribution of patents that the inventor filed by event. Specifically, if the number of newly filed patents for an inventor is among the top 5% in each event sample, this inventor is defined as a superstar inventor. *Superstar* is a dummy variable indicating if the inventor is a superstar inventor based on this definition. We interact *Treat*Post* with *Superstar* and include the interaction term in the baseline regression model. We do not include *Superstar* as an independent variable because it is time-variant, and thus its effects are absorbed by the inventor fixed effects.

The results are reported in Table 5. Columns (1-3) in Table 5 show the estimation of the impact of sports inspiration on superstar inventor innovation productivity. The coefficients of *Treat*Post*Superstar* are negative and significant at the 1% level in all three regressions, implying that relative to more experienced inventors, sports inspiration has a greater effect on the productivity of less experienced inventors/ non-superstar inventors, by not only generating more patents, but more valuable patents.

[Insert Table 5 here]

5.3.2 Underdog Inspiration and Innovation Outcomes

In Table 6, we investigate whether underdog narratives in sports victories inspire more innovation than expected victories. To categorize underdog vs. non-underdog wins, we use preseason betting odds for each championship team. For example, the Cincinnati Reds' 1990 World Series win was deemed an underdog victory since their pre-season odds of winning were +1600, much higher than their competitor, the Oakland Athletics, at +600. In contrast, the Boston Red Sox's 2004 World Series win was considered a non-underdog victory since their pre-season odds of +400 were less than their competitor, the St. Louis Cardinals, at +1500.

By comparing innovation outcomes following underdog versus expected victories, we can isolate the inspirational effect of underdog narratives. Our analysis controls for other factors that may drive innovation, like team popularity and media attention. The results provide useful insights for finance professionals and firms interested in quantifying different sources of inspiration for creativity and innovation. Examining underdog effects contributes to our understanding of how storytelling and framing of competition can motivate performance. Columns (1) to (3) and Columns (4) to (6) in Table 6 show the regression results of the influence of sports inspiration caused by underdog victories and non-underdog victories separately. The coefficients of *Treat*Post* from Column (1) to Column (3) are significantly positive at the 1% level. But the coefficients of *Treat*Post* from Column (4) to Column (6) are not significant, implying that the effects of sports inspiration are more significant on inventor productivity when the sports inspiration is caused by underdog victories.

[Insert Table 6 here]

5.4 The Mechanisms

Our baseline results show that sports inspiration leads to inventors being more innovative. In this section, we investigate the specific channels through which sports inspiration affects inventor productivity.

5.4.1 Sports Inspiration and Inventor Innovation Strategies

The innovation process is unavoidably associated with risk, and there are significant variations in risk-taking across the various innovation strategies. Several studies note that innovation involves the exploration and exploitation of existing projects. (March, 1991; Benner and Tushman, 2002; Balsmeier et al., 2017). March (1991) argues that exploration is characterized by search, variation, risk-taking, experimentation, play, flexibility, discovery, and innovation, while exploitation is characterized by refinement, choice production, efficiency, selection, implementation, and execution. In other words, an exploratory innovation strategy is associated with higher risk than an exploitative innovation strategy (March 1991; Chava et al., 2013; Balsmeier et al., 2017). If being inspired by sports wins increases the risk-taking incentive of inventors, we anticipate that they will be more inclined to venture into new fields after such inspiration. The enhancement in inventor productivity should therefore come primarily from exploratory innovation, rather than exploitative innovation.

By looking at specific variables, we can understand how inventors choose between exploring new ideas or exploiting existing ones in their innovation strategies. We follow Benner and Tushman (2002) and calculate the number of exploratory and exploitative patents filed in a given year. To classify a patent as exploitative, we analyze if over 60% of its backward citations are from the inventor's existing knowledge, which includes their patents or citations made by those patents in the past five years. The Variable *Exploit* is the adjusted number of exploitative patents filed by each inventor in each year. *Explore* measures the amount of patents filed by each inventor that have more than 60% of their citations outside of the inventor's knowledge pool. It is worth noting that multiple inventors are often registered for one patent and one patent could be exploitative to one inventor but exploratory to another.

We re-estimate the baseline regression model using these two variables as our dependent variables. The results are presented in Table 7. Column (1) shows that the coefficient of *Treat*Post* is positive and statistically significant when *Explore*_{*t*+*t*} is the dependent variable. This result indicates that the treated inventors increased their efforts to explore unfamiliar fields after the sports inspiration. Column (2) shows that the coefficient of *Treat*Post* is positive and unsignificant when *Exploit*_{*t*+1} is the dependent variable. Even though the coefficient of *Treat*Post* in Column (2) is positively significant, which suggests that the treated inventors produce more exploratory patents than exploitative patents after the sports inspiration relative to control inventors.

Overall, the results suggest that after sports inspiration events, the treated inventors are more likely to explore new and unfamiliar fields of research. Given that an exploratory innovation strategy involves more risk than an exploitative innovation strategy, the finding provides supporting evidence for the risk-taking channel. In other words, sports inspiration encourages risk-taking behavior and results in greater inventor productivity.

[Insert Table 7 here]

5.4.2 Sports Inspiration and Average Patent Quality

Our baseline results show that the treated inventors produce more patents after sports inspiration. These patents also generate more forward citations and have higher economic value relative to inventors who are not affected by the sports inspiration. To examine whether the increase in innovation output by treated inventors is at least partly driven by improved cognitive function (which affects inventors' R&D decisions), we examine whether the average

quality of each patent among treated inventors improves after the sports inspiration. Average higher patent quality indicates improved R&D capability and improved cognitive abilities of these inventors.

To perform this test, we adopt two average patent quality measures. The average number of citations per patent (*LnAvgCit*) is defined as the natural logarithm of 1 plus the average number of forward citations received by an inventor's newly filed patents. Average economic value per patent (*LnAvgVal*) is defined as the natural logarithm of 1 plus the average economic value of an inventor's newly filed patents. We first calculate the two variables across all newly filed patents for each inventor. We also calculate the two variables across new exploratory patents and new exploitative patents separately. In addition, we calculate the two variables across cooperative patents and sole patents as well.

We re-estimate the baseline regression model using two average patent quality measures as dependent variables. The regression results are presented in Table 8. Column (1) and Column (2) report the results of all newly filed patents. The coefficients of *Treat*Post* are positive and statistically significant in both regressions, suggesting that patents generated by treated inventors have a higher average number of citations and higher economic value after the sports inspiration events. As such, sports inspiration enhances not only the number but also the average quality of patents. Both factors contribute to the increase in total patent citation and economic value.

Next, we examine the exploratory and exploitative, separately from Column (3) to Column (6). We can observe that the coefficient of Treat*Post in Column (3) is positive and statistically significant at 1% significance level. But the coefficients of Treat*Post in Column (5) and Column (6) are positive but not statistically significant at 5% significance level, which is also consistent with what we found in Section 5.4.1. Collectively, the findings in this section suggest that improved cognitive performance serves as a supplementary channel through which sports inspiration increases innovation.

[Insert Table 8 here]

5.5 Further Analysis

In our baseline analysis, we focus on inventors affiliated with publicly listed firms so that we can control for innovation inputs and the characteristics of the firms in which the inventors work. In this section, we extend the analysis to all inventors regardless of whether they are affiliated with publicly listed firms or not. As such, the sample in this test includes all U.S. inventors based on the treatment and control group we defined previously. We control for Ln_tenure and Event_Inventor, Event_City, and Event_Year fixed effects in the regression model (1) using a stacked DiD model. The regression results are presented in Table 10.

Column (1) and Column (2) in Table 9 show the estimation of the impact of sports inspiration on the number of patents and citations for all inventors. The coefficients of Treat*Post are positive and significant at the 1% level (coefficient = 0.010 with standard errors = 0.004) and 1% level (coefficient = 0.010 with standard errors = 0.004) respectively. Column (3) and Column (4) in Table 5 show the estimation of the impact of sports inspiration on the quality of patents (citation) for all inventors. The coefficients of Treat*Post are positive and significant at the 1% level (coefficient = 0.036 with standard errors = 0.013) and 1% level (coefficient = 0.036 with standard errors = 0.013) and 1% level (coefficient = 0.036 with standard errors = 0.013) respectively. The results are consistent with our baseline results, which suggests that our finding holds for all inventors, not just those affiliated with publicly listed firms.

[Insert Table 9 here]

In addition, considering the real economic effect of sports inspiration, we aggregate the innovation activities from the inventor level to the city level to explore whether sports inspiration could stimulate innovation activities for the entire city. We construct two dependent variables: *City_Num*, which equals the natural logarithm of one plus the number of newly filed patents in a city in a given year, and *City_Cit*, which equals the natural logarithm of one plus the number of one plus the number of forward citations of newly filed patents in a city in a given year. Columns (1) and (2) in Table 10 show the estimation of the impact of sports inspiration on the number of patents and citations for the city. The coefficients of Treat*Post are positive and significant at the 1% level (coefficient = 1.070 with standard errors = 0.251) and the 1% level (coefficient = 1.228 with standard errors = 0.398), respectively. The results suggest that sports inspiration not only stimulates innovation activities at the inventor level but also at the city level.

[Insert Table 10 here]

6. Conclusion

Our study reveals a powerful link between sports inspiration and inventor performance. Local sports victories serve as an unexpected catalyst, boosting inventors' productivity and creativity. The impact is clear: more patents filed, higher forward citations, and increased economic value of innovations. Importantly, inspired inventors don't just produce more—they aim higher, targeting newer, more challenging technologies. This demonstrates how positive emotions from sports triumphs can fuel workplace creativity and focus. For firms and managers, these findings are a game-changer. They highlight the untapped potential of positive affect and inspirational narratives in driving innovation. By harnessing the power of inspiration, companies could unlock significant productivity gains and breakthrough innovations. Our research opens new avenues for motivating inventor performance, suggesting that the path to corporate innovation might just run through the sports arena.

Reference

- Acharya, V. V., Baghai, R. P., & Subramanian, K. V. (2014). Wrongful discharge laws and innovation. The Review of Financial Studies, 27(1), 301-346.
- Aghion, P., Van Reenen, J., & Zingales, L. (2013). Innovation and institutional ownership. American Economic Review, 103(1), 277-304.
- Akhigbe, A., Newman, M., & Whyte, A. M. (2017). Predictable sports sentiment and local trading. Financial Management, 46(2), 433-453.
- Arkes, H. R., Herren, L. T., & Isen, A. M. (1988). The role of potential loss in the influence of affect on risk-taking behavior. Organizational Behavior and Human Decision Processes, 42(2), 181-193.
- Ashton, J. K., Gerrard, B., & Hudson, R. (2003). Economic impact of national sporting success: evidence from the London stock exchange. Applied Economics Letters, 10(12), 783-785.
- Baghai, R., Silva, R., & Ye, L. (2019). Teams and bankruptcy. Swedish House of Finance Research Paper, (17-19).
- Baker, A. C., Larcker, D. F., & Wang, C. C. (2022). How much should we trust staggered difference-in-differences estimates? Journal of Financial Economics, 144(2), 370-395.
- Baas, M., De Dreu, C. K., & Nijstad, B. A. (2008). A meta-analysis of 25 years of moodcreativity research: Hedonic tone, activation, or regulatory focus? Psychological Bulletin, 134(6), 779.
- Bassi, A., Colacito, R., & Fulghieri, P. (2013). 'O sole mio: An experimental analysis of weather and risk attitudes in financial decisions. The Review of Financial Studies, 26(7), 1824-1852.
- Balkin, D. B., Markman, G. D., & Gomez-Mejia, L. R. (2000). Is CEO pay in high-technology firms related to innovation? Academy of Management Journal, 43(6), 1118-1129.
- Balsmeier, B., Fleming, L., & Manso, G. (2017). Independent boards and innovation. Journal of Financial Economics, 123(3), 536-557.
- Benner, M. J., & Tushman, M. (2002). Process management and technological innovation: A longitudinal study of the photography and paint industries. Administrative Science Quarterly, 47(4), 676-707.
- Cardazzi, A., McCannon, B. C., Humphreys, B. R., & Rodriguez, Z. (2022). Emotional cues and violent behavior: Unexpected basketball losses increase incidents of family violence. The Journal of Law, Economics, and Organization, ewac014.
- Cengiz, D., Dube, A., Lindner, A., & Zipperer, B. (2019). The effect of minimum wages on low-wage jobs. The Quarterly Journal of Economics, 134(3), 1405-1454.
- Chang, S. C., Chen, S. S., Chou, R. K., & Lin, Y. H. (2012). Local sports sentiment and returns of locally headquartered stocks: A firm-level analysis. Journal of Empirical Finance, 19(3), 309-318.
- Chang, X., Fu, K., Low, A., & Zhang, W. (2015). Non-executive employee stock options and corporate innovation. Journal of Financial Economics, 115(1), 168-188.
- Chava, S., Oettl, A., Subramanian, A., & Subramanian, K. V. (2013). Banking deregulation and innovation. Journal of Financial Economics, 109(3), 759-774.
- Chen, C., Chen, Y., Hsu, P. H., & Podolski, E. J. (2016). Be nice to your innovators: Employee treatment and corporate innovation performance. Journal of Corporate Finance, 39, 78-98.
- Chen, Y., Hsu, P. H., Podolski, E., & Veeraraghavan, M. (2022). In the Mood for Creativity: Weather-Induced Mood, Inventor Performance, and Firm Value. Inventor Performance, and Firm Value (November 25, 2022).

- Chen, Y., Podolski, E. J., Rhee, S. G., & Veeraraghavan, M. (2014). Local gambling preferences and corporate innovative success. Journal of Financial and Quantitative Analysis, 49(1), 77-106.
- Cohen, W. M., Nelson, R., & Walsh, J. P. (2000). Protecting their intellectual assets: Appropriability conditions and why US manufacturing firms patent (or not).
- Davis, M. A. (2009). Understanding the relationship between mood and creativity: A metaanalysis. Organizational Behavior and Human Decision Processes, 108(1), 25-38.
- Depetris-Chauvin, E., Durante, R., & Campante, F. (2020). Building nations through shared experiences: Evidence from African football. American Economic Review, 110(5), 1572-1602.
- Ederer, F., & Manso, G. (2011). Incentives for innovation: Bankruptcy, corporate governance, and compensation systems. Handbook of Law, Innovation, and Growth, 90-111.
- Ederer, F., & Manso, G. (2013). Is pay for performance detrimental to innovation? Management Science, 59(7), 1496-1513.
- Edmans, A., Garcia, D., & Norli, Ø. (2007). Sports sentiment and stock returns. The Journal of Finance, 62(4), 1967-1998.
- Eisenhardt, K. M., & Martin, J. A. (2000). Dynamic capabilities: what are they? Strategic Management Journal, 21(10-11), 1105-1121.
- Fich, E. M., Nguyen, T., & Petmezas, D. (2023). The effects of terrorist attacks on inventor productivity and mobility. Research Policy, 52(1), 104655.
- Fredrickson, B. L. (1998). What good are positive emotions? Review of General Psychology, 2(3), 300-319.
- Galasso, A., & Simcoe, T. S. (2011). CEO overconfidence and innovation. Management Science, 57(8), 1469-1484.
- Gao, H., Hsu, P. H., & Zhang, J. (2023). Pay Transparency and Inventor Productivity: Evidence from State-level Pay Secrecy Laws. Available at SSRN 3632849.
- Gao, H., Hsu, P. H., Li, K., & Zhang, J. (2020). The real effect of smoking bans: Evidence from corporate innovation. Journal of Financial and Quantitative Analysis, 55(2), 387-427.
- Ganco, M., Ziedonis, R. H., & Agarwal, R. (2015). More stars stay, but the brightest ones still leave: Job hopping in the shadow of patent enforcement. Strategic Management Journal, 36(5), 659-685.
- Ge, Q., Sarmiento Barbieri, I., & Schneider, R. (2021). Sporting events, emotional cues, and crime: Spatial and temporal evidence from brazilian soccer games. Economic Inquiry, 59(1), 375-395.
- Griliches, Z. (Ed.). (1987). R&D, patents and productivity. University of Chicago Press.
- Hall, B. H. (1993). The stock market's valuation of R&D investment during the 1980's. The American Economic Review, 83(2), 259-264.
- Hall, B. H. (2002). The financing of research and development. Oxford Review of Economic Policy, 18(1), 35-51.
- Hall, B., Helmers, C., Rogers, M., & Sena, V. (2014). The choice between formal and informal intellectual property: a review. Journal of Economic Literature, 52(2), 375-423.
- Hall, B. H., Jaffe, A., & Trajtenberg, M. (2005). Market value and patent citations. RAND Journal of Economics, 16-38.
- Hall, B. H., Jaffe, A. B., & Trajtenberg, M. (2001). The NBER patent citation data file: Lessons, insights and methodological tools.
- Hall, B. H., & Ziedonis, R. H. (2001). The patent paradox revisited: an empirical study of patenting in the US semiconductor industry, 1979-1995. RAND Journal of Economics, 101-128.

- Hall, R. (1993). A framework linking intangible resources and capabiliites to sustainable competitive advantage. Strategic Management Journal, 14(8), 607-618.
- Henderson, R., & Cockburn, I. (1994). Measuring competence? Exploring firm effects in pharmaceutical research. Strategic Management Journal, 15(S1), 63-84.
- Hirshleifer, D., Low, A., & Teoh, S. H. (2012). Are overconfident CEOs better innovators?. The Journal of Finance, 67(4), 1457-1498.
- Hirt, E. R., Zillmann, D., Erickson, G. A., & Kennedy, C. (1992). Costs and benefits of allegiance: Changes in fans' self-ascribed competencies after team victory versus defeat. Journal of Personality and Social Psychology, 63(5), 724.
- Hombert, J., & Matray, A. (2017). The real effects of lending relationships on innovative firms and inventor mobility. The Review of Financial Studies, 30(7), 2413-2445.
- Isen, A. M. (1999). Positive affect. Handbook of Cognition and Emotion, 20, 522-539.
- Isen, A. M. (2000). Positive effect and decision making. Handbook of Emotions, 2, 417-435.
- Isen, A. M., Johnson, M., Mertz, E., & Robinson, G. F. (1985). The influence of positive affect on the unusualness of word associations. Journal of Personality and Social Psychology, 48(6), 1413.
- Isen, A. M., Shalker, T. E., Clark, M., & Karp, L. (1978). Affect, accessibility of material in memory, and behavior: A cognitive loop?. Journal of Personality and Social Psychology, 36(1), 1.
- Jaffe, A. B., & De Rassenfosse, G. (2017). Patent citation data in social science research: Overview and best practices. Journal of the Association for Information Science and Technology, 68(6), 1360-1374.
- Jamison, K. R. (1989). Mood disorders and patterns of creativity in British writers and artists. Psychiatry, 52(2), 125.
- Jones, M. V., Coffee, P., Sheffield, D., Yangüez, M., & Barker, J. B. (2012). Just a game? Changes in English and Spanish soccer fans' emotions in the 2010 World Cup. Psychology of Sport and Exercise, 13(2), 162-169.
- Kaplanski, G., & Levy, H. (2010). Exploitable predictable irrationality: The FIFA World Cup effect on the U.S. stock market. Journal of Financial and Quantitative Analysis, 45(2), 535-553.
- Kaplanski, G., Levy, H., Veld, C., & Veld-Merkoulova, Y. (2015). Do happy people make optimistic investors? Journal of Financial and Quantitative Analysis, 50(1-2), 145-168.
- Kavanagh, D. J., & Bower, G. H. (1985). Mood and self-efficacy: Impact of joy and sadness on perceived capabilities. Cognitive Therapy and Research, 9(5), 507-525.
- Kogan, L., Papanikolaou, D., Seru, A., & Stoffman, N. (2017). Technological innovation, resource allocation, and growth. The Quarterly Journal of Economics, 132(2), 665-712.
- Lam, A. (2011). What motivates academic scientists to engage in research commercialization: 'Gold', 'ribbon'or 'puzzle'?. Research Policy, 40(10), 1354-1368.
- Lerner, J., & Seru, A. (2022). The use and misuse of patent data: Issues for finance and beyond. The Review of Financial Studies, 35(6), 2667-2704.
- Lerner, J., & Wulf, J. (2007). Innovation and incentives: Evidence from corporate R&D. the Review of Economics and Statistics, 89(4), 634-644.
- Li, G. C., Lai, R., D'Amour, A., Doolin, D. M., Sun, Y., Torvik, V. I., & Fleming, L. (2014). Disambiguation and co-authorship networks of the US patent inventor database (1975– 2010). Research Policy, 43(6), 941-955.
- Luo, Y., Chen, Y., & Lin, J. C. (2022). Does air quality affect inventor productivity? Evidence from the NOx budget program. Journal of Corporate Finance, 73, 102170.
- March, J. G. (1991). Exploration and exploitation in organizational learning. Organization Science, 2(1), 71-87.
- Mitton, T. (2021). Economic significance in corporate finance. Available at SSRN 3667830.

- Moretti, E. (2021). The effect of high-tech clusters on the productivity of top inventors. American Economic Review, 111(10), 3328-3375.
- Osterloh, M., & Frey, B. S. (2000). Motivation, knowledge transfer, and organizational forms. Organization Science, 11(5), 538-550.
- Pantzalis, C., & Park, J. C. (2014). Exuberance out of left field: Do sports results cause investors to take their eyes off the ball? Journal of Economic Behavior & Organization, 107, 760-780.
- Petersen, M. A. (2008). Estimating standard errors in finance panel data sets: Comparing approaches. The Review of Financial Studies, 22(1), 435-480.
- Porter, M. E. (1992). Capital choices: Changing the way America invests in industry. Journal of Applied Corporate Finance, 5(2), 4-16.
- Rosenbaum, P. R., & Rubin, D. B. (1983). The central role of the propensity score in observational studies for causal effects. Biometrika, 70(1), 41-55.
- Sauermann, H., & Cohen, W. M. (2010). What makes them tick? Employee motives and firm innovation. Management Science, 56(12), 2134-2153.
- Sauermann, H., & Roach, M. (2014). Not all scientists pay to be scientists: PhDs' preferences for publishing in industrial employment. Research Policy, 43(1), 32-47.
- Schwarz, N., Strack, F., Kommer, D., & Wagner, D. (1987). Soccer, rooms, and the quality of your life: Mood effects on judgments of satisfaction with life in general and with specific domains. European Journal of Social Psychology, 17(1), 69-79.
- Schweitzer, K., Zillmann, D., Weaver, J. B., & Luttrell, E. S. (1992). Profile: Perception of threatening events in the emotional aftermath of a televised college football game. Journal of Broadcasting & Electronic Media, 36(1), 75-82.
- Solow, R. M. (1957). Technical change and the aggregate production function. The Review of Economics and Statistics, 39(3), 312-320.
- Staw, B. M. (1995). Why no one really wants creativity. Creative Action in Organizations, 161, 66.
- Stern, S. (2004). Do scientists pay to be scientists? Management Science, 50(6), 835-853.
- To, M. L., Fisher, C. D., Ashkanasy, N. M., & Rowe, P. A. (2012). Within-person relationships between mood and creativity. Journal of Applied Psychology, 97(3), 599.
- Tversky, A., & Kahneman, D. (1973). Availability: A heuristic for judging frequency and probability. Cognitive Psychology, 5(2), 207-232.
- Wann, D. L., Dolan, T. J., MeGeorge, K. K., & Allison, J. A. (1994). Relationships between spectator identification and spectators' perceptions of influence, spectators' emotions, and competition outcome. Journal of Sport and Exercise Psychology, 16(4), 347-364.
- White, H. (1980). A heteroskedasticity-consistent covariance matrix estimator and a direct test for heteroskedasticity. Econometrica: journal of the Econometric Society, 817-838.
- Zacchia, P. (2018). Benefiting colleagues but not the city: Localized effects from the relocation of superstar inventors. Research Policy, 47(5), 992-1005.
- Zivin, J. G., & Neidell, M. (2012). The impact of pollution on worker productivity. American Economic Review, 102(7), 3652-3673.

Table 1. Summary statistics

This table reports the descriptive statistics of the sample we used in our main analysis. The sample consists of 277,022 inventor-year observations for 30 events from 1975 to 2007, obtained from the HBS patent inventor database and Compoutat. Appendix A provides a detailed description of the construction of the variables. All the continuous variables are winsorized at the 1% level.

	Obs.	Mean	Median	Std.	Min	Max
LnPat	297203	0.244	0.000	0.367	0	1.649
LnCit	297203	1.022	0.000	1.428	0	5.136
LnVal	297203	0.857	0.000	1.211	0	4.668
Asset	297203	8.652	8.867	1.952	3.157	12.178
R&D	297203	0.08	0.069	0.065	0	0.373
Cash	297203	0.186	0.113	0.192	0.002	0.81
ROA	297203	0.134	0.150	0.117	-0.406	0.361
CapEx	297203	0.066	0.059	0.043	0.006	0.22
Leverage	297203	0.185	0.169	0.147	0	0.655
Tobin's Q	297203	2.615	1.910	2.222	0.855	14.404
Firm_age	297203	3.164	3.497	0.817	0.693	4.007
Ln_tenure	297203	1.907	1.946	0.815	0	3.332

Table 2. Baseline Result

This table reports the baseline results of the influence of sports inspiration on inventor productivity using the stacked DID model. $LnPat_{t+1}$ is the natural log of the sum of the adjusted patent numbers for an inventor in a specific year, which is used to measure the quantity of an inventor's productivity. Columns (1) and (2) show the results of sports inspiration on $LnPat_{t+1}$. $LnCit_{t+1}$ and $LnVal_{t+1}$ are the natural log of the sum of the adjusted patent forward citations and value for an inventor in a specific year respectively, which are used to measure the quality of an inventor's productivity. Columns (3) and (6) show the results of sports inspiration on $LnCit_{t+1}$ and $LnVal_{t+1}$, respectively. Continuous variables are winsorized at the 1% level. Appendix A provides a detailed description of the construction of the variables. Standard errors are in parentheses and are clustered at the event-city level. ***, **, and * indicate the significance at the 1%, 5%, and 10% levels, respectively.

Variable	(1)	(2)	(3)	(4)	(5)	(6)
	$LnPat_{t+1}$	$LnPat_{t+1}$	$LnCit_{t+1}$	$LnCit_{t+1}$	$LnVal_{t+1}$	$LnVal_{t+1}$
Treat*Post	0.021***	0.021***	0.098***	0.098***	0.103***	0.098^{***}
	(0.007)	(0.007)	(0.024)	(0.024)	(0.027)	(0.026)
Asset		0.013***		0.057^{***}	~ /	0.076^{***}
		(0.004)		(0.017)		(0.016)
R&D		-0.003		-0.253		0.218
		(0.045)		(0.171)		(0.141)
Cash		0.015		0.075		0.001
		(0.013)		(0.058)		(0.050)
ROA		0.001		0.028		0.227^{***}
		(0.019)		(0.069)		(0.053)
CapEx		0.118***		0.498^{***}		0.751^{***}
		(0.037)		(0.154)		(0.127)
Leverage		-0.038***		-0.155***		-0.035
		(0.013)		(0.054)		(0.041)
Tobin's Q		0.003***		0.012^{***}		0.010^{***}
		(0.001)		(0.002)		(0.003)
Firm_age		-0.023*		-0.127***		-0.243***
		(0.012)		(0.048)		(0.038)
Ln_tenure		0.011**		0.038^{**}		0.030^{**}
		(0.004)		(0.017)		(0.014)
Constant	0.244^{***}	0.173^{***}	1.018^{***}	0.818^{***}	0.853^{***}	0.792***
	(0.000)	(0.049)	(0.001)	(0.210)	(0.001)	(0.192)
Event_Inventor	YES	YES	YES	YES	YES	YES
Event_City	YES	YES	YES	YES	YES	YES
Event_Year	YES	YES	YES	YES	YES	YES
Event_Firm	YES	YES	YES	YES	YES	YES
Obs.	292879	292879	292879	292879	292879	292879
adj. R^2	0.373	0.373	0.309	0.310	0.331	0.332

Table 3. Dynamic Effect

This table reports the dynamic effect of the influence of sports inspiration on inventor productivity. Columns (1) to (3) show the dynamic effect of sports inspiration on $LnPat_{t+1}$, $LnCit_{t+1}$, and $LnVal_{t+1}$, respectively. Continuous variables are winsorized at the 1% level. Appendix A provides a detailed description of the construction of the variables. Standard errors are in parentheses and are clustered at the event-city level. ***, **, and * indicate the significance at the 1%, 5%, and 10% levels, respectively.

Variable	(1)	(2)	(3)
	$LnPat_{t+1}$	$LnCit_{t+1}$	$LnVal_{t+1}$
Treat*Before ²	-0.002	-0.030	0.001
	(0.008)	(0.047)	(0.031)
Treat*Before ¹	0.012	-0.006	0.030
	(0.011)	(0.059)	(0.034)
Treat*After ¹	0.024^{*}	0.074	0.094^{**}
	(0.014)	(0.052)	(0.044)
Treat*After ²	0.012	0.042	0.104***
	(0.013)	(0.060)	(0.036)
Treat*After ³	0.027^{**}	0.153**	0.130**
	(0.012)	(0.060)	(0.052)
Asset	0.013***	0.054***	0.076^{***}
	(0.005)	(0.021)	(0.018)
R&D	0.017	-0.297	0.239
	(0.051)	(0.185)	(0.154)
Cash	0.001	0.001	-0.044
	(0.016)	(0.064)	(0.059)
ROA	-0.005	0.024	0.258^{***}
	(0.022)	(0.081)	(0.062)
CapEx	0.139***	0.574^{***}	0.842***
	(0.046)	(0.177)	(0.149)
Leverage	-0.044***	-0.159**	-0.051
	(0.013)	(0.063)	(0.045)
Tobin's Q	0.003***	0.014***	0.011***
	(0.001)	(0.003)	(0.003)
Firm_age	-0.022	-0.136**	-0.229***
	(0.014)	(0.054)	(0.043)
Ln_tenure	0.007	0.022	0.012
	(0.005)	(0.018)	(0.015)
Constant	0.179***	0.911***	0.769^{***}
	(0.052)	(0.235)	(0.198)
Event_Inventor	YES	YES	YES
Event_City	YES	YES	YES
Event_Year	YES	YES	YES
Event_Firm	YES	YES	YES
Obs.	235324	235324	235324
adj. R^2	0.368	0.307	0.331

Table 4. Robustness tests

This table reports a series of robustness checks in our project. Columns (1) to (3) show the influence of sports inspiration on $LnPat_{t+1}$, $LnCit_{t+1}$, and $LnVal_{t+1}$, respectively, when we delete the inventors who do not file any patents during the sample period we use. Columns (4) to (6) show the influence of sports inspiration on $LnPat_{t+1}$, $LnCit_{t+1}$, and $LnVal_{t+1}$, respectively, when we delete the inventors who are affiliated with utility and financial firms. Columns (7) to (9) show the influence of sports inspiration on $LnPat_{t+1}$, $LnCit_{t+1}$, and $LnVal_{t+1}$, respectively, when we change the control group including the inventors who are from the same state as the treated inventors to that including the inventors who live in the city which never win a champion of big four professional sports in U.S. in the U.S. Columns (10) to (12) show the influence of sports inspiration on $LnPat_{t+1}$, $LnCit_{t+1}$, and $LnVal_{t+1}$, respectively, when we use PSM method to construct a balanced sample. Continuous variables are winsorized at the 1% level. Appendix A provides a detailed description of the variables. Standard errors are in parentheses and are clustered at the event-city level. ***, **, and * indicate the significance at the 1%, 5%, and 10% levels, respectively.

	R	Robustness check	1	R	obustness check	2]	Robustness check	3	R	obustness check	4
Variable	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)
	$LnPat_{t+1}$	$LnCit_{t+1}$	$LnVal_{t+1}$	$LnPat_{t+1}$	$LnCit_{t+1}$	$LnVal_{t+1}$	$LnPat_{t+1}$	$LnCit_{t+1}$	$LnVal_{t+1}$	$LnPat_{t+1}$	$LnCit_{t+1}$	$LnVal_{t+1}$
Treat*Post	0.025^{***}	0.117^{***}	0.121***	0.020^{***}	0.097^{***}	0.096***	0.019***	0.078^{***}	0.093***	0.028^{***}	0.111^{***}	0.081^{**}
	(0.008)	(0.029)	(0.034)	(0.007)	(0.024)	(0.026)	(0.007)	(0.022)	(0.029)	(0.009)	(0.036)	(0.032)
Asset	0.014^{***}	0.059^{***}	0.079^{***}	0.014^{***}	0.063***	0.080^{***}	0.019^{***}	0.075^{***}	0.079^{***}	0.026^{**}	0.099	0.115***
	(0.004)	(0.019)	(0.017)	(0.004)	(0.017)	(0.016)	(0.001)	(0.004)	(0.004)	(0.011)	(0.066)	(0.044)
R&D	0.002	-0.255	0.256^{*}	-0.001	-0.235	0.227	0.052^{***}	0.277^{***}	0.215***	0.104	0.617	0.629
	(0.047)	(0.179)	(0.147)	(0.045)	(0.171)	(0.141)	(0.014)	(0.055)	(0.052)	(0.168)	(0.863)	(0.696)
Cash	0.017	0.082	-0.001	0.024^{*}	0.091	0.007	0.040^{***}	0.098^{***}	0.080^{***}	-0.016	-0.207	-0.292**
	(0.015)	(0.062)	(0.055)	(0.013)	(0.057)	(0.051)	(0.005)	(0.017)	(0.016)	(0.040)	(0.181)	(0.144)
ROA	0.000	0.030	0.249***	-0.003	0.023	0.229^{***}	0.012^{**}	0.106^{***}	0.161***	0.033	-0.026	0.050
	(0.021)	(0.075)	(0.057)	(0.019)	(0.070)	(0.054)	(0.005)	(0.018)	(0.016)	(0.084)	(0.389)	(0.287)
CapEx	0.134***	0.564^{***}	0.843^{***}	0.118^{***}	0.482^{***}	0.750^{***}	0.182^{***}	0.602^{***}	0.538^{***}	0.266^{**}	0.464	0.654^{*}
	(0.042)	(0.171)	(0.141)	(0.037)	(0.154)	(0.128)	(0.011)	(0.038)	(0.031)	(0.108)	(0.453)	(0.365)
Leverage	-0.044***	-0.179***	-0.041	-0.035***	-0.144***	-0.023	-0.007^{*}	0.031**	0.074^{***}	-0.023	-0.206	-0.230
	(0.014)	(0.061)	(0.045)	(0.013)	(0.055)	(0.042)	(0.004)	(0.015)	(0.014)	(0.040)	(0.219)	(0.157)
Tobin's Q	0.003***	0.013***	0.010^{***}	0.003***	0.012^{***}	0.010^{***}	0.004^{***}	0.019^{***}	0.019^{***}	0.007	0.059^{**}	0.060***
	(0.001)	(0.003)	(0.003)	(0.001)	(0.003)	(0.003)	(0.000)	(0.001)	(0.001)	(0.005)	(0.029)	(0.021)
Firm_age	-0.016	-0.079^{**}	-0.169***	-0.026**	-0.129***	-0.248***	-0.034***	-0.164***	-0.261***	0.060	0.309	-0.115
	(0.010)	(0.040)	(0.030)	(0.012)	(0.049)	(0.038)	(0.003)	(0.010)	(0.009)	(0.040)	(0.189)	(0.093)
Ln_tenure	0.011^{**}	0.041**	0.029^{*}	0.010^{**}	0.036**	0.028^{**}	0.009^{***}	0.019^{***}	0.015^{***}	0.010	0.066	0.040
	(0.005)	(0.019)	(0.016)	(0.004)	(0.017)	(0.014)	(0.001)	(0.003)	(0.003)	(0.009)	(0.041)	(0.034)
Constant	0.172^{***}	0.774^{***}	0.632***	0.171^{***}	0.773***	0.771^{***}	0.112^{***}	0.610^{***}	0.776^{***}	-0.305*	-1.349*	-0.110
	(0.047)	(0.206)	(0.186)	(0.049)	(0.211)	(0.194)	(0.016)	(0.053)	(0.051)	(0.158)	(0.782)	(0.517)
Event_Inventor	YES	YES	YES	YES	YES	YES	YES	YES	YES	YES	YES	YES
Event_City	YES	YES	YES	YES	YES	YES	YES	YES	YES	YES	YES	YES
Event_Year	YES	YES	YES	YES	YES	YES	YES	YES	YES	YES	YES	YES
Event_Firm	YES	YES	YES	YES	YES	YES	YES	YES	YES	YES	YES	YES
Obs.	258216	258216	258216	290522	290522	290522	6042173	6042173	6042173	24622	24622	24622
adj. <i>R</i> ²	0.337	0.262	0.288	0.371	0.308	0.331	0.358	0.302	0.340	0.287	0.226	0.267

Table 5. Subsample Analysis: Superstar inventors or not

This table reports the influence of sports inspiration on inventor productivity based on inventor characteristics (superstar inventor or not). Columns (1) to (3) show the results of the sports inspiration on superstar inventors' innovation productivity when including the interaction term *Treat*Post*Superstar*. Continuous variables are winsorized at the 1% level. Appendix A provides a detailed description of the construction of the variables. Standard errors are in parentheses and are clustered at the event-city level. ***, **, and * indicate the significance at the 1%, 5%, and 10% levels, respectively.

Variable	(1)	(2)	(3)
	$LnPat_{t+1}$	$LnCit_{t+1}$	$LnVal_{t+1}$
Treat*Post	0.032***	0.135***	0.124***
	(0.006)	(0.022)	(0.023)
Treat* Post*Superstar	-0.303***	-0.972***	-0.678***
*	(0.037)	(0.122)	(0.171)
Asset	0.013***	0.057^{***}	0.076***
	(0.004)	(0.017)	(0.016)
R&D	-0.003	-0.251	0.219
	(0.045)	(0.171)	(0.141)
Cash	0.016	0.076	0.002
	(0.013)	(0.058)	(0.050)
ROA	0.001	0.027	0.226***
	(0.019)	(0.069)	(0.053)
CapEx	0.119***	0.500^{***}	0.752^{***}
•	(0.037)	(0.154)	(0.127)
Leverage	-0.038***	-0.155***	-0.035
	(0.013)	(0.054)	(0.041)
Tobin's Q	0.003***	0.012^{***}	0.010^{***}
	(0.001)	(0.002)	(0.003)
Firm_age	-0.023*	-0.128***	-0.243***
	(0.012)	(0.048)	(0.038)
Ln_tenure	0.011^{**}	0.038^{**}	0.029^{**}
	(0.004)	(0.017)	(0.014)
Constant	0.173***	0.818^{***}	0.792***
	(0.048)	(0.209)	(0.192)
Event_Inventor	YES	YES	YES
Event_City	YES	YES	YES
Event_Year	YES	YES	YES
Event_Firm	YES	YES	YES
Obs.	292879	292879	292879
adj. R^2	0.373	0.310	0.332

Table 6. Subsample Analysis: Underdog match or not

This table shows the results of the influence of sports inspiration caused by underdog matches or non-underdog matches on inventor productivity. Columns (1) to (3) show the results of the sports inspiration caused by underdog matches on inventors' productivity. Columns (4) to (6) show the results of the sports inspiration caused by non-underdog matches on inventor productivity. Continuous variables are winsorized at the 1% level. Appendix A provides a detailed description of the construction of the variables. Standard errors are in parentheses and are clustered at the event-city level. ***, **, and * indicate the significance at the 1%, 5%, and 10% levels, respectively.

	Underdog Match			Non-underdog Match			
Variable	(1)	(2)	(3)	(4)	(5)	(6)	
	$LnPat_{t+1}$	$LnCit_{t+1}$	$LnVal_{t+1}$	$LnPat_{t+1}$	$LnCit_{t+1}$	$LnVal_{t+1}$	
Treat*Post	0.018^{***}	0.090^{***}	0.097^{***}	0.020	0.085	0.102	
	(0.006)	(0.021)	(0.031)	(0.025)	(0.088)	(0.080)	
Asset	0.023***	0.105^{***}	0.159^{***}	0.008	0.067^{**}	0.042	
	(0.007)	(0.029)	(0.023)	(0.008)	(0.034)	(0.033)	
R&D	-0.015	-0.355	0.529^{***}	-0.118	-0.452	-0.073	
	(0.077)	(0.287)	(0.196)	(0.093)	(0.421)	(0.383)	
Cash	0.017	0.064	-0.030	0.092^{***}	0.389^{**}	0.379^{**}	
	(0.019)	(0.083)	(0.064)	(0.036)	(0.171)	(0.154)	
ROA	-0.034	-0.112	0.128^*	-0.013	-0.248^{*}	-0.187	
	(0.028)	(0.093)	(0.071)	(0.035)	(0.147)	(0.123)	
CapEx	0.137**	0.666^{***}	0.591***	0.027	0.100	0.234	
	(0.059)	(0.214)	(0.168)	(0.064)	(0.322)	(0.248)	
Leverage	-0.023	-0.086	-0.031	-0.086***	-0.353***	-0.166**	
	(0.018)	(0.079)	(0.056)	(0.023)	(0.103)	(0.077)	
Tobin's Q	0.004^{***}	0.018^{***}	0.019^{***}	0.003	0.012	0.008	
	(0.001)	(0.004)	(0.004)	(0.002)	(0.010)	(0.009)	
Firm_age	-0.002	-0.008	-0.121***	-0.028	-0.098	-0.276***	
	(0.011)	(0.037)	(0.032)	(0.022)	(0.107)	(0.073)	
Ln_tenure	0.026***	0.095^{***}	0.083***	-0.001	-0.003	-0.022	
	(0.005)	(0.022)	(0.017)	(0.007)	(0.024)	(0.022)	
Constant	-0.008	-0.089	-0.443**	0.243^{**}	0.724	1.378^{***}	
	(0.064)	(0.280)	(0.218)	(0.108)	(0.520)	(0.449)	
Event_Inventor	YES	YES	YES	YES	YES	YES	
Event_City	YES	YES	YES	YES	YES	YES	
Event_Year	YES	YES	YES	YES	YES	YES	
Event_Firm	YES	YES	YES	YES	YES	YES	
Obs.	140089	140089	140089	76964	76964	76964	
adj. R^2	0.377	0.307	0.330	0.334	0.276	0.324	

Table 7. Mechanism I: Innovation Strategy (Exploration and Exploitation)

This table presents the results of the influence of sports inspiration on inventor innovation strategy. Column (1) shows the results of the influence of sports inspiration on the exploratory patents. Column (2) shows the results of the influence of sports inspiration on the exploitative. Continuous variables are winsorized at the 1% level. Appendix A provides a detailed description of the construction of the variables. Standard errors are in parentheses and are clustered at the event-city level. ***, **, and * indicate the significance at the 1%, 5%, and 10% levels, respectively.

Variable	(1)	(2)
	Exploration	Exploitation
	$Explore_{t+1}$	$Exploit_{t+1}$
Treat*Post	0.013**	0.006
	(0.006)	(0.004)
Asset	0.006^{*}	0.006***
	(0.004)	(0.002)
R&D	-0.037	0.020
	(0.037)	(0.021)
Cash	0.000	0.020***
	(0.011)	(0.008)
ROA	-0.012	-0.001
	(0.017)	(0.009)
CapEx	0.064**	0.049***
*	(0.032)	(0.017)
Leverage	-0.021*	-0.019***
0	(0.011)	(0.006)
Tobin's Q	0.004^{***}	-0.001**
~	(0.001)	(0.000)
Firm_age	-0.040***	0.022***
_ 0	(0.010)	(0.006)
Ln_tenure	-0.006	0.017***
—	(0.004)	(0.002)
Constant	0.259***	-0.102***
	(0.040)	(0.026)
Event_Inventor	YES	YES
Event_City	YES	YES
Event_Year	YES	YES
Event_Firm	YES	YES
Obs.	292512	292512
adj. R ²	0.273	0.305

Table 8. Mechanism II: Average patent quality

This table presents the regression results of the influence of sports inspiration on average patent quality, measured by the average citations and economic value per patent. Continuous variables are winsorized at the 1% level. Appendix A provides a detailed description of the construction of the variables. Standard errors are in parentheses and are clustered at the event-city level. ***, **, and * indicate the significance at the 1%, 5%, and 10% levels, respectively.

	All p	atents	Explorati	ve patents	Exploitat	ive patents
Variable	(1)	(2)	(3)	(4)	(5)	(6)
	LnAvgCit _{t+1}	LnAvgVal _{t+1}	LnAvgCit _{t+1}	LnAvgVal _{t+1}	LnAvgCit _{t+1}	LnAvgVal _{t+1}
Treat*Post	0.099^{***}	0.099^{***}	0.072^{***}	0.041	0.017	0.031*
	(0.028)	(0.030)	(0.028)	(0.025)	(0.018)	(0.019)
Asset	0.076***	0.105^{***}	0.041^{**}	0.051***	0.043***	0.060^{***}
	(0.019)	(0.017)	(0.017)	(0.016)	(0.011)	(0.011)
R&D	-0.322^{*}	0.254	-0.358**	0.128	0.002	0.014
	(0.176)	(0.156)	(0.173)	(0.146)	(0.135)	(0.103)
Cash	0.089	0.055	0.070	0.049	0.087^{*}	0.067^{*}
	(0.065)	(0.061)	(0.059)	(0.052)	(0.044)	(0.038)
ROA	0.074	0.312***	-0.040	0.178^{***}	0.057	0.071^{*}
	(0.069)	(0.056)	(0.070)	(0.059)	(0.048)	(0.040)
CapEx	0.427^{**}	0.737***	0.225	0.449^{***}	0.287^{**}	0.421***
	(0.169)	(0.149)	(0.149)	(0.131)	(0.117)	(0.092)
Leverage	-0.139**	0.011	-0.035	0.040	-0.123***	-0.052^{*}
	(0.063)	(0.048)	(0.053)	(0.044)	(0.041)	(0.029)
Tobin's Q	0.012^{***}	0.007^{**}	0.017^{***}	0.017^{***}	-0.004*	-0.011***
	(0.003)	(0.003)	(0.003)	(0.002)	(0.002)	(0.002)
Firm_age	-0.160***	-0.349***	-0.274***	-0.315***	0.073^{**}	-0.022
	(0.048)	(0.040)	(0.043)	(0.036)	(0.034)	(0.024)
Ln_tenure	0.040^{**}	0.038^{***}	-0.023	-0.037***	0.114***	0.107^{***}
	(0.017)	(0.014)	(0.016)	(0.013)	(0.013)	(0.012)
Constant	0.933***	1.010^{***}	1.451***	1.288^{***}	-0.472***	-0.343**
	(0.231)	(0.207)	(0.207)	(0.193)	(0.155)	(0.134)
Event_Inventor	YES	YES	YES	YES	YES	YES
Event_City	YES	YES	YES	YES	YES	YES
Event_Year	YES	YES	YES	YES	YES	YES
Event_Firm	YES	YES	YES	YES	YES	YES
Obs.	292512	292512	292512	292512	292512	292512
adj. <i>R</i> ²	0.241	0.275	0.168	0.190	0.257	0.247

Table 9. Further Analysis: All Inventors

This table shows the result of the influence of sports inspiration on all inventors' innovation productivity. Continuous variables are winsorized at the 1% level. Appendix A provides a detailed description of the construction of the variables. Standard errors are in parentheses and are clustered at the event-city level. ***, **, and * indicate the significance at the 1%, 5%, and 10% levels, respectively.

Variable	(1)	(2)	(3)	(4)
	$LnPat_{t+1}$	$LnPat_{t+1}$	$LnCit_{t+1}$	$LnCit_{t+1}$
Treat*Post	0.010^{**}	0.010^{**}	0.036***	0.036***
	(0.004)	(0.004)	(0.013)	(0.013)
Ln_tenure		0.008^{***}		0.035^{***}
		(0.002)		(0.007)
Constant	0.229^{***}	0.215^{***}	0.573^{***}	0.506^{***}
	(0.000)	(0.004)	(0.000)	(0.014)
Event_Inventor	YES	YES	YES	YES
Event_City	YES	YES	YES	YES
Event_Year	YES	YES	YES	YES
Obs.	826955	826955	826955	826955
adj. R ²	0.349	0.349	0.309	0.309

Table 10. Further Analysis: City level innovation activities

This table shows the result of the influence of sports inspiration on city-level innovation productivity. Continuous variables are winsorized at the 1% level.. Standard errors are in parentheses and are clustered at the event-city level. ***, **, and * indicate the significance at the 1%, 5%, and 10% levels, respectively.

Variable	(1)	(2)
	$City_Num_{t+1}$	$City_Cit_{t+1}$
Treat*Post	1.070^{***}	1.228^{***}
	(0.251)	(0.398)
Constant	0.779^{***}	1.516^{***}
	(0.000)	(0.001)
Event_City	YES	YES
Event_Year	YES	YES
Obs.	67460	67460
adj. \mathbb{R}^2	0.670	0.570

Variable	Definition
Measure of Innovation Output	
LnPat	Natural logarithm of one plus the number of newly filed patents. Note: for each newly filed patent, we scale it by the total number of inventors of this patent
LnCit	Natural logarithm of one plus the number of forward citations received by the newly filed patents. Note: for the forward citation of each newly filed patent, we scale it by the total number of inventors of this patent
LnVal	Natural logarithm of one plus the total economic value of newly filed patents. Note: for the economic value of each newly filed patent, we scale it by the total number of inventors of this patent
Explore	The number of newly filed patents for which more than 60% of backward citations are outside of the inventor's knowledge base. Note: for each newly filed patent, we scale it by the total number of inventors of this patent
Exploit	The number of newly filed patents for which more than 60% of backward citations are within the inventor's knowledge base. Note: for each newly filed patent, we scale it by the total number of inventors of this patent
LnAvgCit	Natural logarithm of one plus the average of forward citations received by the newly filed patents. Note: for each newly filed patent and the forward citation of each newly filed patent, we scale it by the total number of inventors of this patent
LnAvgVal	Natural logarithm of one plus the average economic value of the newly filed patents. Note: for each newly filed patent and for the economic value of each newly filed patent, we scale it by the total number of inventors of this patent
Firm Characteristics	
Asset	Natural logarithm of total assets.
R&D	R&D expenditures normalized by total assets.
Cash	Cash and short-term investments normalized by total assets.
ROA	Net income normalized by total assets.
CapEx	Capital expenditure normalized by total assets.
Leverage Tobin's Q	Total debt normalized by total assets. Market value of equity plus the book value of total assets minus the book value of equity minus balance sheet deferred taxes, normalized by the book value of total assets.
Firm_age	Natural logarithm of one plus the number of years since a firm's first appearance in the Compustat database.

Appendix A: Variable Definitions

Inventor Characteristics

Ln_tenure

Natural logarithm of one plus the number of years between the year that the inventor enters the patent database and the observation year.

Appendix B. Additional Tests

Considering the intra-cluster correlation, many patents are assigned at the team/firm level, which means there is a strong (mechanical) correlation in patent output across inventors within a team-firm. This paper focuses on the individual inventor level; therefore, we re-run the baseline analysis clustered at the event-firm level to rule out this concern.

This table reports the baseline results of the influence of sports inspiration on inventor productivity using the stacked DID model. $LnPat_{t+1}$ is the natural log of the sum of the adjusted patent numbers for an inventor in a specific year, which is used to measure the quantity of an inventor's productivity. Columns (1) and (2) show the results of sports inspiration on $LnPat_{t+1}$. $LnCit_{t+1}$ and $LnVal_{t+1}$ are the natural log of the sum of the adjusted patent forward citations and value for an inventor in a specific year respectively, which are used to measure the quality of an inventor's productivity. Columns (3) and (6) show the results of sports inspiration on $LnCit_{t+1}$ and $LnVal_{t+1}$, respectively. Continuous variables are winsorized at the 1% level. Appendix A provides a detailed description of the construction of the variables. Standard errors are in parentheses and are clustered at the event-firm level. ***, **, and * indicate the significance at the 1%, 5%, and 10% levels, respectively.

Variable	(1)	(2)	(3)	(4)	(5)	(6)
	$LnPat_{t+1}$	$LnPat_{t+1}$	$LnCit_{t+1}$	$LnCit_{t+1}$	$LnVal_{t+1}$	$LnVal_{t+1}$
Treat*Post	0.021***	0.021***	0.098***	0.098***	0.103***	0.098^{***}
	(0.007)	(0.007)	(0.024)	(0.024)	(0.027)	(0.026)
Asset	. ,	0.013***	. ,	0.057^{***}		0.076***
		(0.004)		(0.017)		(0.016)
R&D		-0.003		-0.253		0.218
		(0.045)		(0.171)		(0.141)
Cash		0.015		0.075		0.001
		(0.013)		(0.058)		(0.050)
ROA		0.001		0.028		0.227^{***}
		(0.019)		(0.069)		(0.053)
CapEx		0.118***		0.498^{***}		0.751***
		(0.037)		(0.154)		(0.127)
Leverage		-0.038***		-0.155***		-0.035
		(0.013)		(0.054)		(0.041)
Tobin's Q		0.003^{***}		0.012***		0.010***
		(0.001)		(0.002)		(0.003)
Firm_age		-0.023^{*}		-0.127***		-0.243***
		(0.012)		(0.048)		(0.038)
Ln_tenure		0.011^{**}		0.038^{**}		0.030^{**}
		(0.004)		(0.017)		(0.014)
Constant	0.244^{***}	0.173***	1.018^{***}	0.818***	0.853^{***}	0.792^{***}
	(0.000)	(0.049)	(0.001)	(0.210)	(0.001)	(0.192)
Event_Inventor	YES	YES	YES	YES	YES	YES
Event_City	YES	YES	YES	YES	YES	YES
Event_Year	YES	YES	YES	YES	YES	YES
Event_Firm	YES	YES	YES	YES	YES	YES
Obs.	292879	292879	292879	292879	292879	292879
adj. R^2	0.373	0.373	0.309	0.310	0.331	0.332